Author's response to reviews

Title: Low bone mineral density is a significant risk factor for low-energy distal radius fractures in middle-aged and elderly men: A case-control study

Authors:

Jannike Øyen (Jannike.Oyen@isf.uib.no)
Gudrun Rohde (gudrun.elin.rohde@sshf.no)
Marc Hochberg (mhocber@umaryland.edu)
Villy Johnsen (villy.johnsen@sshf.no)
Glenn Haugeberg (glenn.haugeberg@sshf.no)

Version: 5 Date: 27 March 2011

Author's response to reviews:

1. The , behind glenn Haugeberg has been delete.
2. 16.000 is now written as 16,000
3. Section Authors contribution: the abbreviation for GER is changed to GR
4. Reference 7 has been updated with correct citation, this article has now been published.